Author's response to reviews

Title: Can achievement goal theory provide a useful motivational perspective for explaining psychosocial attributes of medical students?

Authors:

Nir Madjar (madjarn.bgu@gmail.com)
Bachner G Yaacov (bachner@bgu.ac.il)
Talma Kushnir (talmak@bgu.ac.il)

Version: 4 Date: 12 January 2012

Author's response to reviews:

Final submission after further formatting changes.